Author's response to reviews

Title: Discontinuation of hypnotics during cognitive behavioural therapy for insomnia

Authors:

Lucie Zavesicka (zavesicka@pcp.lf3.cuni.cz)
Martin Brunovsky (brunovsky@pcp.lf3.cuni.cz)
Milos Matousek (matousek@pcp.lf3.cuni.cz)
Peter Sos (sos@pcp.lf3.cuni.cz)

Version: 7 Date: 17 September 2008

Author's response to reviews: see over
Authors’ letter to the BioMed Central Editorial Production Team

2008-09-15

Title: Discontinuation of hypnotics during cognitive behavioural therapy for insomnia

Version: 5 Date: 2008-09-15

Dear BioMed Central Editorial Production Team,

Thank you for providing review of our manuscript “Discontinuation of hypnotics during cognitive behavioural therapy for insomnia.” We have revised our manuscript and have made all requested formatting changes.

Regarding the individual points:

Formatting changes requested
1. Minor revisions - we will appreciate your help with these changes as you propose.

2. Please remove table placement text - we have removed table placement texts in all manuscript

3. Authors' contributions - Please indicate that all authors read and approved the final manuscript. We have done it according to your instruction.

4. Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance. We have carefully re-read manuscript and have made final corrections. We believe that we now have addressed all of the comments and suggestions and hope that the manuscript now is acceptable for publication.

Yours, with best wishes,

Lucie Zavesicka, Martin Brunovsky, Milos Matousek, Peter Sos